Author’s response to reviews

Title: Effect of Intra-Pregnancy Nonsurgical Periodontal Therapy on Inflammatory Biomarkers and Adverse Pregnancy Outcomes: A Systematic Review with Meta-Analysis.

Authors:

HELBERT EUSTÁQUIO CARDOSO DA SILVA (helbertcardososilva@gmail.com)
Cristine Stefani (cmstefani@gmail.com)
Nilce Melo (nilce@unb.br)
Adriano de Lima (adadlima@gmail.com)
Cassiano Rösing (ckrosing@hotmail.com)
André Porporatti (andreporporatti@yahoo.com.br)
Graziela Canto (graziela.canto@ufsc.br)

Version: 3 Date: 04 Sep 2017

Author’s response to reviews:

Responses to the Editors

SYSR-D-17-00092


HELBERT EUSTÁQUIO CARDOSO DA SILVA, M.D.; Cristine Miron Stefani, PhD; Nilce de Santos Melo, PhD; Adriano de Almeida de Lima, PhD; Cassiano Kuchenbecker Rösing, PhD; André Luís Porporatti, PhD; Graziela De Lucca Canto, PhD

This manuscript is ready to be accepted, but the authors need to make the same change regarding the p = 0.07 and p =0.007 in the abstract that they made in the text.

The change in abstract was made according to suggest from reviewers